Author's response to reviews

Title: A counterbalanced cross-over study of the effects of visual, auditory and no feedback on performance measures in a simulated cardiopulmonary resuscitation

Authors:

Carolyn L Cason (clcason@uta.edu)
Cynthia Trowbridge (ctrowbridge@uta.edu)
Susan M Baxley (sbaxley@uta.edu)
Mark D Ricard (ricard@uta.edu)

Version: 7 Date: 1 August 2011

Author's response to reviews: see over
Changes to main document

A period was added to follow ‘et al’----‘et al.’ on the following pages/lines

P3L20, P4L4, P5L5, P5L9, P6L10, P6L11, P6L21, P13L17